OXFORD MATHEMATICAL MONOGRAPHS

Series Editors

J. M. BALL E. M. FRIEDLANDER W. T. GOWERS N. J. HITCHIN I. G. MACDONALD L. NIRENBERG R. PENROSE J. T. STUART

OXFORD MATHEMATICAL MONOGRAPHS

A. Belleni-Moranti: Applied semigroups and evolution equations

A.M. Arthurs: Complementary variational principles 2nd edition

M. Rosenblum and J. Rovnyak: Hardy classes and operator theory

J.W.P. Hirschfeld: Finite projective spaces of three dimensions

A. Pressley and G. Segal: Loop groups

D.E. Edmunds and W.D. Evans: Spectral theory and differential operators

Wang Jianhua: The theory of games

S. Omatu and J.H. Seinfeld: Distributed parameter systems: theory and applications

J. Hilgert, K.H. Hofmann, and J.D. Lawson: Lie groups, convex cones, and semigroups

S. Dineen: The schwarz lemma

S.K. Donaldson and P.B. Kronheimer: The geometry of four-manifolds

D.W. Robinson: Elliptic operators and Lie groups

A.G. Werschulz: The computational complexity of differential and integral equations

L. Evens: Cohomology of groups

G. Effinger and D.R. Hayes: Additive number theory of polynomials

J.W.P. Hirschfeld and J.A. Thas: General Galois geometries

P.N. Hoffman and J.F. Humpherys: Projective representations of the symmetric groups

1. Györi and G. Ladas: The oscillation theory of delay differential equations

J. Heinonen, T. Kilpelainen, and O. Martio: Non-linear potential theory

B. Amberg, S. Franciosi, and F. de Giovanni: Products of groups

M.E. Gurtin: Thermomechanics of evolving phase boundaries in the plane

I. Ionescu and M. Sofonea: Functional and numerical methods in viscoplasticity

N. Woodhouse: Geometric quantization 2nd edition

U. Grenander: General pattern theory

J. Faraut and A. Koranyi: Analysis on symmetric cones

I.G. Macdonald: Symmetric functions and Hall polynomials 2nd edition

B.L.R. Shawyer and B.B. Watson: Borel's methods of summability

M. Holschneider: Wavelets: an analysis tool

Jacques Thévenaz: G-algebras and modular representation theory

Hans-Joachim Baues: Homotopy type and homology P.D.D'Eath: Black holes: gravitational interactions

R. Lowen: Approach spaces: the missing link in the topology-uniformity-metric traid

Nguyen Dinh Cong: Topological dynamics of random dynamical systems J.W.P. Hirschfeld: Projective geometries over finite fields 2nd edition

K. Matsuzaki and M. Taniguchi: Hyperbolic manifolds and Kleinian groups

David E. Evans and Yasuyuki Kawahigashi: Quantum symmetries on operator algebras

Norbert Klingen: Arithmetical similarities: prime decomposition and finite group theory

Isabelle Catto, Claude Le Bris, and Pierre-Louis Lions: The mathematical theory of thermo-

dynamic limits: Thomas-Fermi type models

D. McDuff and D. Salamon: Introduction to symplectic topology 2nd edition

William M. Goldman: Complex hyberbolic geometry

Charles J. Colbourn and Alexander Rosa: Triple systems

V.A. Kozlov, V.G. Maz'ya and A.B. Movchan: Asymptotic analysis of fields in multi-structures

Gérard A. Maugin: Nonlinear waves in elastic crystals

George Dassios and Ralph Kleinman: Low frequency scattering

Gerald W. Johnsor and Michel L. Lapidus: The Feynman integral and

Fevnman's operational calculus

W. Lay and S.Y. Slavyanov: Special functions: A unified theory based on singularities

D. Joyce: Compact manifolds with special holonomy

A. Carbone and S. Semmes: A graphic apology for symmetry and implicitness

Johann Boos: Classical and modern methods in summability

Nigel Higson and John Roe: Analytic K-Homology

S. Semmes: Some novel types of fractal geometry

Tadeusz Iwaniec and Gaven Martin: Geometric function theory and nonlinear analysis

The Feynman Integral and Feynman's Operational Calculus

Gerald W. Johnson

Department of Mathematics and Statistics University of Nebraska-Lincoln

Michel L. Lapidus

Department of Mathematics University of California, Riverside This book has been printed digitally and produced in a standard specification in order to ensure its continuing availability

OXFORD UNIVERSITY PRESS

Great Clarendon Street, Oxford OX2 6DP

Oxford University Press is a department of the University of Oxford. It furthers the University's objective of excellence in research, scholarship, and education by publishing worldwide in

Oxford New York

Auckland Cape Town Dar es Salaam Hong Kong Karachi Kuala Lumpur Madrid Melbourne Mexico City Nairobi New Delhi Shanghai Taipei Toronto With offices in

Argentina Austria Brazil Chile Czech Republic France Greece Guatemala Hungary Italy Japan South Korea Poland Portugal Singapore Switzerland Thailand Turkey Ukraine Vietnam

Oxford is a registered trade mark of Oxford University Press in the UK and in certain other countries

Published in the United States by Oxford University Press Inc., New York

© G. W. Johnson and M. L. Lapidus

The moral rights of the author have been asserted

Database right Oxford University Press (maker)

Reprinted 2010

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without the prior permission in writing of Oxford University Press, or as expressly permitted by law, or under terms agreed with the appropriate reprographics rights organization. Enquiries concerning reproduction outside the scope of the above should be sent to the Rights Department, Oxford University Press, at the address above

You must not circulate this book in any other binding or cover And you must impose this same condition on any acquirer

ISBN 978-0-19-851572-2

To Joan, my love and best friend, and to our family: Tom. Lisa. Caitlin. Carly and Hannah: Greg. Melissa and Sarah: Katie: Jenny and Doug

Gerald W. Johnson

To my parents, Myriam and Serge Lapidus

> To my wife and love, Odile Lapidus

To my children, Julie and Michaël

To my sisters and their families, Sylvie Hanus and Muriel Attia

For their love, teaching, encouragement, and inspiration

Michel L. Lapidus

PREFACE

This book is directed primarily to mathematicians and mathematical physicists, but also to theoretical physicists and to other scientists with an interest in quantum theory. One of our purposes in writing this book on the beautiful and closely related topics of the Feynman integral and Feynman's operational calculus is to make these subjects accessible to a wider audience, including graduate students. Accordingly, much of the necessary background material is provided within: we call the reader's attention especially to Chapters 3, 4, 6, 9 and 10 in the table of contents. Chapter 7 also consists, in a sense, of background material, but it deals with the heuristic ideas that led to the Feynman integral and with the difficulties that arise from attempts to make this subject mathematically rigorous. Of course, many potential readers will know a significant portion of the background information and will therefore be able to go quickly over the corresponding parts of the book.

Both authors have taught courses in Lincoln and Riverside, respectively, over the material of this book as it was being developed and refined. Also, both of us have given lectures, sometimes series of lectures (or short courses), on these subjects in many places around the world. Our experience suggests that it takes about three to four semesters to go through Chapters 1 to 19. The material divides rather naturally into Chapters 1–13 and Chapters 14–19, although there is a great deal of overlap and cross-referencing between these two parts of the book. Many of the listeners at the courses or conference talks mentioned above have been helpful to us by asking thoughtful questions or by making insightful comments, and perhaps most useful of all, by helping us to maintain our enthusiasm for this long-term endeavor.

Gerald W. Johnson and Michel L. Lapidus.

May 1999

ACKNOWLEDGEMENTS

(joint and individual)

The strong scientific influence of the physicist Richard Feynman will be apparent to the readers of this book, especially in the discussion of his imaginative heuristic ideas in Chapters 7 and 14 in connection with the Feynman integral and Feynman's operational calculus, respectively.

Mathematicians and mathematical physicists whose work has had a major impact on several aspects of this book include Robert Cameron, Mark Kac, Tosio Kato and Edward Nelson.

Acknowledgements by Gerald W. Johnson

I have been fortunate enough to have many positive influences in my scientific life, but it is probably not reasonable to name them all. The following people have been mentors and/or collaborators as well as friends over an extended period: David Skoug, my long-term colleague and collaborator at the University of Nebraska at Lincoln (UNL); Robert Cameron and David Storvick: Gopinath Kallianpur; Kun Soo Chang; Jésus Gil de Lamadrid, my Ph.D. advisor at the University of Minnesota and mentor even long after I had switched areas.

The following Ph.D. students and postdoctoral visitors at UNL have helped with the proofreading of the manuscript and sometimes with useful comments regarding the presentation of the material: Byung Moo Ahn, Lisa Johnson, Jeong Gyoo Kim, Jung Won Ko, Jung Ah Lim, Lance Nielsen, Yeon-Hee Park, Tristan Reyes and Troy Riggs.

Finally, I would like to acknowledge the support of several institutions and individuals going back to the years when this "book" was only a vague plan and some lecture notes from graduate classes at UNL and from talks given elsewhere (needless to say, this plan was later substantially modified and merged with that of my co-author): University of Erlangen, Germany (Dietrich Kölzow); University of North Carolina, Chapel Hill, Center for Stochastic Processes (Gopinath Kallianpur); University of Sherbrooke, Québec (Pedro Morales), University of Clermont-Ferrand, France (Albert Badrikian), Yonsei University, Seoul, Korea (Kun Soo Chang); University of Minnesota (Robert Cameron, David Storvick and Jésus Gil de Lamadrid); University of Georgia, Athens (my co-author); Bielefeld-Bochum Stochastic in Bielefeld, Germany (Sergio Albeverio, Ph. Blanchard, Ludwig Streit); University of Missouri, Columbia (Brian DeFacio); University of California, Riverside (my co-author); University of New South Wales, Sydney, Australia (Brian Jefferies); University of Warsaw, Poland (W. Chojnacki); Institute for Applied Mathematics, Chinese Academy of Sciences, Beijing (Zhiming Ma, J. A. Yan); and finally, of course. University of Nebraska-Lincoln (David Skoug).

Gerald W. Johnson, University of Nebraska-Lincoln

Acknowledgements by Michel L. Lapidus

I would like to acknowledge my indebtedness to the following mathematicians, mathematical physicists and physicists who have played an important role in my formative years as a research mathematician.

Gustave Choquet, for opening up for me the wonderful world of analysis and geometry during my student years in Paris and for remaining a close advisor and gentle critic throughout my scientific life.

Yvonne Choquet-Bruhat, who has directed my very first research memoir (dealing with general relativity and differential topology) and who has provided me with kind guidance and direction at a time of personal hardship.

Haïm Brézis, for guiding some of my first steps into research and for being my advisor both for my Ph.D. Dissertation [La1], my Thèse de Doctorat d'Etat ès Sciences [La13], and my Habilitation (to direct research), all at the Université Pierre et Marie Curie (Paris VI). He has set high ethical and scientific standards by his own example which I have always tried to emulate.

Tosio Kato and Paul Chernoff, for warmly welcoming me at the University of California. Berkeley, and for providing guidance and inspiration during my first few years of academic life in the United States.

Isadore Singer, whose weekly three hour long Gauge Theory (and Mathematical Physics) Seminar at UC Berkeley in the late 1970s and the early 1980s has been an enthralling intellectual and cultural experience for me, in particular by transcending the traditional divides between mathematics and physics and by helping to build a very useful dictionary between these two disciplines.

Mark Kac, of whom I was fortunate to be a junior colleague during the first half of the 1980s in Los Angeles and who has been for me both a close friend and advisor as well as a scientific mentor until his death towards the end of 1984.

Richard Feynman, whom I have had the privilege to know at the California Institute of Technology in Pasadena from 1981 until shortly before his death in 1988. I am grateful to him for listening to some of my ideas and results on the 'Feynman integral' (even by attending, apparently for the first time in Caltech, a Mathematics Colloquium which I gave on this subject in 1982), as well as for suggesting to me to read his 1951 paper on the operational calculus [Fey8] and for strongly encouraging me to develop a mathematical theory justifying and extending it.

Alain Connes, for the depth and intrinsic beauty of his work which I have watched develop since my student years, as well as for many friendly and enlightening conversations about much of mathematics over the last few years.

My long-time friend and mentor, Moshe Flato, who passed away a few months ago and to whom I would have so much liked to give one of the first copies of this book. From him. I have learned more physics and mathematical physics than from anybody else. I will always remember Moshe fondly for his wit, breadth and generosity.

To all of them, as well as to many colleagues and researchers throughout the world with whom I have had the pleasure to collaborate or to interact. I wish to pay homage and give my heartfelt thanks.

In addition, I am grateful to a number of graduate students for attending my lectures or seminars on or related to various parts of the material presented in this book and for providing me with their feedback and comments. Among them, I would like to mention, in particular, Christina He. Derek Dreier, Cheryl Griffith, Piotr Hebda. Peiqing Jiang, Lior Kadosh, Sasa Kresic-Juric, Luong Le, Kathy Nabours and Trieu Nguyen.

On the more personal side, I would like to thank my wife, Odile, and my children, Julie and Michaël, for their patience and understanding, and above all, for not voicing too loudly their doubts and concerns at times when they would have been fully justified to do so. Finishing up within the same year two books on completely different subjects (the present one, which has been a more than ten year enterprise, and the research monograph [La-vF2], which is the realization of another one of my long-term dreams) has certainly not been easy for me, but it was clearly even less so for them who had to maintain their support while watching in disbelief. I promise them not to begin writing another book... at least for a while.

Finally, it is a pleasure to also acknowledge the support of many universities and research institutes throughout the years during which this book was written or the research leading to it was being developed. In particular, the University of California at Berkeley, the University of Southern California in Los Angeles, the Mathematical Sciences Research Institute (MSRI) in Berkeley (on many occasions from the mid-1980s through the late 1990s), the University of Georgia in Athens (at which I have taught my first graduate course on a very preliminary version of the beginning of this material in the late 1980s), Yale University in New Haven, Connecticut, and the University of California at Riverside. Furthermore, I would like to express my deep gratitude to the Institut des Hautes Etudes Scientifiques (IHES) in Bures-sur-Yvette, near Paris, France (of which I have been a very frequent visitor during the last seven years), the Fields Institute for Research in Mathematical Sciences in Toronto (and previously in Waterloo), Canada, the Erwin Schroedinger International Institute for Mathematical Physics in Vienna, Austria, as well as the Isaac Newton Institute for Mathematical Sciences of the University of Cambridge, England, where some of the later chapters of this book were completed.

Last but not least, I wish to acknowledge the financial support of several research foundations, including especially the U.S. National Science Foundation (NSF) which has supported my research for the past fifteen years and, more recently, the Research Foundation of the Academic Senate of the University of California.

Michel L. Lapidus, University of California, Riverside

Both of us would like to thank Mrs. Jan Carter for an excellent job of typing a difficult manuscript and for remaining in good spirits against all odds.

We would also like to thank the Mathematics and Physical Sciences Editors at Oxford University Press, Elizabeth Johnston, Managing Editor, and Julia Tompson, Development Editor, for helpful advice and encouragement as well as careful handling of the manuscript.

CONTENTS

| 1 | Intr | oduction | |
|---|------|---|----------|
| | 1.1 | General introductory comments | J |
| | | Feynman's path integral | 1 |
| | | Feynman's operational calculus | 2 |
| | | Feynman's operational calculus via the Feynman and | |
| | | Wiener integrals | |
| | | Feynman's operational calculus and evolution equations | - |
| | | Further work on or related to the Feynman integral: Chapter 20 | 8 |
| | 1.2 | Recurring themes and their connections with the Feynman integral and Feynman's operational calculus | 8 |
| | | • | |
| | | Product formulas and applications to the Feynman integral Feynman–Kac formula: Analytic continuation in time and mass | 10 |
| | | The role of operator theory | 10 12 |
| | | Connections between the Feynman–Kac and Trotter product formulas | 13 |
| | | Evolution equations | 13 |
| | | Functions of noncommuting operators | 15 |
| | | Time-ordered perturbation series | 15 |
| | | The use of measures | 16 |
| | 1.3 | Relationship with the motivating physical theories: background and | 10 |
| | 1.5 | quantum-mechanical models | 17 |
| | | Physical background | 17 |
| | | Highly singular potentials | |
| | | Time-dependent potentials | 18 |
| | | Phenomenological models: complex and nonlocal potentials | 19 19 |
| | | Prerequisites, new material, and organization of the book | 21 |
| | | | 21 |
| 2 | | physical phenomenon of Brownian motion | 24 |
| | 2.1 | A brief historical sketch | 24 |
| | 2.2 | Einstein's probabilistic formula | 28 |
| 3 | Wier | ner measure | 31 |
| | 3.1 | There is no reasonable translation invariant measure on Wiener space | 32 |
| | 3.2 | Construction of Wiener measure | 34 |
| | 3.3 | Wiener's integration formula and applications | 42 |
| | | Finitely based functions | 43 |
| | | Applications | 45 |
| | | Axiomatic description of the Wiener process | 51 |
| | 3.4 | Nondifferentiability of Wiener paths | 51 |
| | | d-dimensional Wiener measure and Wiener process | 57 |
| | 3.5 | Appendix: Converse measurability results | 57 |
| | 3.6 | Appendix: $\mathcal{B}(X \times Y) = \mathcal{B}(X) \otimes \mathcal{B}(Y)$ | 60 |
| | | | |

xii CONTENTS

| 4 | Scali | ng in Wiener space and the analytic Feynman integral | 62 |
|---|-------|--|-----|
| | 4.1 | Quadratic variation of Wiener paths | 63 |
| | 4.2 | Scale change in Wiener space | 67 |
| | 4.3 | Translation pathologies | 74 |
| | 4.4 | Scale-invariant measurable functions | 77 |
| | 4.5 | The scalar-valued analytic Feynman integral | 79 |
| | 4.6 | The nonexistence of Feynman's "measure" | 82 |
| | 4.7 | Appendix: Some useful Gaussian-type integrals | 85 |
| | 4.8 | Appendix: Proof of formula (4.2.3a) | 87 |
| 5 | | hastic processes and the Wiener process | 89 |
| | 5.1 | Stochastic processes and probability measures on function spaces | 89 |
| | 5.2 | The Kolmogorov consistency theorem | 90 |
| | 5.3 | Two realizations of the Wiener process | 92 |
| 6 | Qua | ntum dynamics and the Schrödinger equation | 94 |
| | 6.1 | Hamiltonian approach to quantum dynamics | 94 |
| | 6.2 | Transition amplitudes and measurement | 95 |
| | 6.3 | The Heisenberg uncertainty principle | 96 |
| | 6.4 | Hamiltonian for a system of particles | 97 |
| 7 | | Feynman integral: heuristic ideas and mathematical difficulties | 99 |
| | 7.1 | Introduction | 99 |
| | 7.2 | Feynman's formula | 101 |
| | | Connections with classical mechanics: The method of stationary phase | 105 |
| | 7.3 | Heuristic derivation of the Schrödinger equation | 106 |
| | 7.4 | Feynman's approximation formula | 109 |
| | 7.5 | Nelson's approach via the Trotter product formula | 111 |
| | | The Trotter product formula | 114 |
| | 7.6 | The approach via analytic continuation | 115 |
| 8 | Sem | igroups of operators: an informal introduction | 121 |
| 9 | | ar semigroups of operators | 127 |
| | 9.1 | Infinitesimal generator | 127 |
| | | Integral equation | 129 |
| | | Evolution equation | 130 |
| | | Closed unbounded operators | 130 |
| | 9.2 | Examples of semigroups and their generators | 134 |
| | | The translation semigroup | 134 |
| | | The heat semigroup | 135 |
| | | The Poisson semigroup | 137 |
| | 9.3 | The resolvent | 138 |
| | 9.4 | Generation theorems | 140 |
| | | The Hille-Yosida theorem | 140 |
| | | Dissipative operators and the Lumer-Phillips theorem | 141 |
| | 9.5 | Uniformly continuous and weakly continuous semigroups | 143 |
| | 9.6 | Self-adjoint operators, unitary groups and Stone's theorem | 144 |
| | 0.7 | Desturbation theorems | 147 |

CONTENTS xiii

| 10 | Unbo | unded self-adjoint operators and quadratic forms | 152 |
|----|-------|--|-----|
| | 10.1 | Spectral theorem for unbounded self-adjoint operators | 153 |
| | | Multiplication operators | 153 |
| | | Three useful forms of the spectral theorem | 155 |
| | 10.2 | Applications of the spectral theorem | 161 |
| | | The free Hamiltonian H ₀ | 161 |
| | | The heat semigroup and unitary group | 163 |
| | | Standard cores for the free Hamiltonian | 171 |
| | | Imaginary resolvents | 174 |
| | 10.3 | Representation theorems for unbounded quadratic forms | 176 |
| | | Basic definitions and properties | 176 |
| | | Representation theorems for quadratic forms | 183 |
| | | The form sum of operators | 190 |
| | 10.4 | Conditions on the potential V for H_0 -form boundedness | 193 |
| 11 | Produ | uct formulas with applications to the Feynman integral | 197 |
| | 11.1 | Trotter and Chernoff product formulas | 198 |
| | | Product formula for unitary groups | 203 |
| | 11.2 | Feynman integral via the Trotter product formula | 204 |
| | | Criteria for essential self-adjointness of positive operators | 204 |
| | | A brief outline of distribution theory | 205 |
| | | Kato's distributional inequality | 206 |
| | | Essential self-adjointness of the Hamiltonian $H = H_0 + V$ | 210 |
| | | Conditions on the potential V for H_0 -operator boundedness | 215 |
| | | Feynman integral via the Trotter product formula for | 217 |
| | | unitary groups | |
| | 11.3 | Product formula for imaginary resolvents | 220 |
| | | Hypotheses and statement of the main result | 220 |
| | | Proof of the product formula | 222 |
| | | Consequences, extensions and open problems | 229 |
| | 11.4 | Application to the modified Feynman integral | 232 |
| | | Modified Feynman integral and Schrödinger equation with singular potential | 233 |
| | | Extensions: Riemannian manifolds and magnetic vector potentials | 241 |
| | 11.5 | Dominated convergence theorem for the modified Feynman integral | 245 |
| | | Preliminaries | 247 |
| | | Perturbation of form sums of self-adjoint operators | 248 |
| | | Application to a general dominated convergence theorem | 252 |
| | | for Feynman integrals | |
| | 11.6 | The modified Feynman integral for complex potentials | 259 |
| | | Product formula for imaginary resolvents of normal operators | 260 |
| | | Application to dissipative quantum systems | 263 |
| | 11.7 | Appendix: Extended Vitali's theorem with application to unitary groups | 266 |
| | | Extension of Vitali's theorem for sequences of analytic functions | 266 |
| | | Analytic continuation and product formula for unitary groups | 269 |

xiv CONTENTS

| 12 | The F | eynman–Kac formula | 272 |
|----|-------|--|-----|
| | 12.1 | The Feynman-Kac formula, the heat equation and the | |
| | | Wiener integral | 273 |
| | 12.2 | Proof of the Feynman-Kac formula | 276 |
| | | Bounded potentials | 276 |
| | | Monotone convergence theorems for forms and integrals | 282 |
| | | Unbounded potentials | 284 |
| | 12.3 | Consequences | 290 |
| 13 | Analy | tic-in-time or -mass operator-valued Feynman integrals | 293 |
| | 13.1 | Introduction | 293 |
| | 13.2 | The analytic-in-time operator-valued Feynman integral | 298 |
| | 13.3 | Proof of existence | 300 |
| | 13.4 | The Feynman integrals compared with one another and with the | |
| | | unitary group. Application to stability theorems | 303 |
| | 13.5 | The analytic-in-mass operator-valued Feynman integral | 308 |
| | | Definition of the analytic-in-mass operator-valued Feynman integral | 309 |
| | | Nelson's results | 312 |
| | | Haugsby's result for time-dependent, complex-valued potentials | 321 |
| | | Further extensions via a product formula for semigroups | 323 |
| | 13.6 | The analytic-in-mass modified Feynman integral | 331 |
| | | Existence of the analytic-in-mass modified Feynman integral | 334 |
| | | Product formula for resolvents: The case of imaginary mass | 337 |
| | | Comparison with other analytic-in-mass Feynman integrals | 345 |
| | | Highly singular central potentials—the attractive inverse-square potential | 346 |
| | 13.7 | The analytic-in-time operator-valued Feynman integral via additive | |
| | | functionals of Brownian motion | 358 |
| | | Introductory remarks | 359 |
| | | The parallel with Section 13.3 | 359 |
| | | Generalized signed measures | 361 |
| | | The generalized Kato class | 361 |
| | | Capacity on \mathbb{R}^d | 362 |
| | | Smooth measures | 363 |
| | | Positive continuous additive functionals of Brownian motion | 364 |
| | | The relationship between smooth measures and PCAFs | 365 |
| | | The analytic-in-time operator-valued Feynman integral exists for | |
| | | $\mu = \mu_+ - \mu \in S - GK_d$ | 367 |
| | | Examples | 368 |
| 14 | Feyni | nan's operational calculus for noncommuting | |
| | opera | tors: an introduction | 374 |
| | 14.1 | Functions of operators | 375 |
| | 14.2 | The rules for Feynman's operational calculus | 376 |
| | | Feynman's time-ordering convention | 377 |
| | | Feynman's heuristic rules | 377 |
| | | Two elementary examples | 378 |

CONTENTS xv

| | 14.3 | Time-ordered perturbation series | 383 |
|----|-----------|---|-------------|
| | | Perturbation series via Feynman's operational calculus | 383 |
| | | Perturbation series via a path integral | 389 |
| | | The origins of Feynman's operational calculus | 393 |
| | 14.4 | Making Feynman's operational calculus rigorous | 394 |
| | | I. Rigor via path integrals | 394 |
| | | II. Well-defined and useful formulas arrived at via Feynman's | |
| | | heuristic rules | 395 |
| | | III. A general theory of Feynman's operational calculus with | |
| | | computations which are rigorous at each stage | 396 |
| | 14.5 | Feynman's operational calculus via Wiener and Feynman integrals: | |
| | | Comments on Chapters 15–18 | 397 |
| 15 | Gene | ralized Dyson series, the Feynman integral and Feynman's | |
| | opera | tional calculus | 404 |
| | 15.1 | Introduction | 404 |
| | 15.2 | The analytic operator-valued Feynman integral | 407 |
| | | Notation and definitions | 407 |
| | | The analytic (in mass) operator-valued Feynman integral $K'_{\lambda}(\cdot)$ | 410 |
| | | Preliminary results | 413 |
| | 15.3 | A simple generalized Dyson series $(\eta = \mu + \omega \delta_{\tau})$ | 416 |
| | | The classical Dyson series | 424 |
| | 15.4 | Generalized Dyson series: The general case | 426 |
| | 15.5 | Disentangling via perturbation expansions: Examples | 434 |
| | | A single measure and potential | 435 |
| | | Several measures and potentials | 442 |
| | 15.6 | Generalized Feynman diagrams | 446 |
| | 15.7 | Commutative Banach algebras of functionals: | |
| | | The disentangling algebras | 451 |
| | | The disentangling algebras A_i | 452 |
| | | The time-reversal map on A_t and the natural physical ordering | 455 |
| | | Connections with Feynman's operational calculus | 459 |
| 16 | Stabil | ity results | 462 |
| | 16.1 | Stability in the potentials | 462 |
| | 16.2 | Stability in the measures | 464 |
| 17 | The F | eynman-Kac formula with a Lebesgue-Stieltjes | |
| | | re and Feynman's operational calculus | 47 7 |
| | 17.1 | Introduction | 477 |
| | - · · · · | Notation and hypotheses | 478 |
| | | A. E | |

xvi CONTENTS

| | 17.2 | The Feynman-Kac formula with a Lebesgue-Stieltjes measure: Finitely supported discrete part ν | 480 |
|----|-------|---|-----|
| | | Integral equation (integrated form of the evolution equation) | 480 |
| | | Differential equation (differential form of the evolution equation) | 481 |
| | | Discontinuities (in time) of the solution | 482 |
| | | Propagator and explicit solution | 483 |
| | 17.3 | Derivation of the integral equation in a simple case $(\eta = \mu + \omega \delta_{\tau})$ | 486 |
| | | Sketch of the proof when v is finitely supported | 495 |
| | 17.4 | Discontinuities of the solution to the evolution equation | 496 |
| | | The time discontinuities | 496 |
| | | Differential equation and change of initial condition | 497 |
| | 17.5 | Explicit solution and physical interpretations | 499 |
| | | Continuous measure: Uniqueness of the solution | 500 |
| | | Measure with finitely supported discrete part: Propagator and explicit | |
| | | solution | 501 |
| | | Physical interpretations in the quantum-mechanical case | 503 |
| | | Physical interpretations in the diffusion case | 504 |
| | | Further connections with Feynman's operational calculus | 505 |
| | 17.6 | The Feynman-Kac formula with a Lebesgue-Stieltjes measure: The general | |
| | | case (arbitrary measure η) | 507 |
| | | Integral equation (integrated form of the evolution equation) | 507 |
| | | Basic properties of the solution to the integral equation | 509 |
| | | Quantum-mechanical case: Reformulation in the interaction | |
| | | (or Dirac) picture | 511 |
| | | Product integral representation of the solution | 514 |
| | | Distributional differential equation (true differential form of the | |
| | | evolution equation) | 517 |
| | | Unitary propagators | 519 |
| | | Scattering matrix and improper product integral | 520 |
| | | Sketch of the proof of the integral equation | 521 |
| 18 | Nonce | ommutative operations on Wiener functionals, disentangling | |
| 10 | | ras and Feynman's operational calculus | 530 |
| | 18.1 | Introduction | 530 |
| | 18.2 | Preliminaries: maps, measures and measurability | 532 |
| | 18.3 | The noncommutative operations * and + | 535 |
| | 18.4 | The functional integrals $K_{\lambda}^{i}(\cdot)$ and the operations * and $+$ | 540 |
| | 18.5 | The disentangling algebras A_i , the operations * and $\dot{+}$, and the | |
| | 10.5 | disentangling process | 544 |
| | | Examples: Trigonometric, binomial and exponential formulas | 552 |
| | 10 6 | Appendix: Quantization, axiomatic Feynman's operational calculus, and | 332 |
| | 18.6 | generalized functional integral | 553 |
| | | Algebraic and analytic axioms | 554 |
| | | Consequences of the axioms | 556 |
| | | Examples: the disentangling algebras and analytic Feynman integrals | 559 |
| | | Examples, the awenungting argeorus and analytic regiment integrals | 237 |

| CONTENTS | XV11 |
|----------|------------|
| | 75 7 1 1 1 |

| 19 | Feynman's operational calculus and evolution equations | | |
|----|--|---|-----|
| | 19.1 | Introduction and hypotheses | 562 |
| | | Feynman's operational calculus as a generalized path integral | 562 |
| | | Exponentials of sums of noncommuting operators | 563 |
| | | Disentangling exponentials of sums via perturbation series | 563 |
| | | Local and nonlocal potentials | 565 |
| | | Hypotheses | 566 |
| | 19.2 | Disentangling $\exp\{-t\alpha + \int_0^t \beta(s)\mu(ds)\}\$ | 568 |
| | 19.3 | Disentangling $\exp\{-t\alpha + \int_0^t \beta_1(s)\mu_1(ds) + \cdots + \int_0^t \beta_n(s)\mu_n(ds)\}$ | 573 |
| | 19.4 | Convergence of the disentangled series | 581 |
| | 19.5 | The evolution equation | 587 |
| | 19.6 | Uniqueness of the solution to the evolution equation | 596 |
| | 19.7 | Further examples of the disentangling process | 599 |
| | | Nonlocal potentials relevant to phenomenological nuclear theory | 604 |
| 20 | Furth | ner work on or related to the Feynman integral | 609 |
| | 20.1 | Transform approaches to the Feynman integral. References to further | |
| | | approaches | 609 |
| | | A. The Fresnel integral and other transform approaches to the | |
| | | Feynman integral | 610 |
| | | The Fresnel integral | 610 |
| | | Properties of the Fresnel integral | 611 |
| | | An approach to the Feynman integral via the Fresnel integral | 613 |
| | | Advantages and disadvantages of Fresnel integral approaches to the | |
| | | Feynman integral | 613 |
| | | The Feynman map | 615 |
| | | The Poisson process and transforms | 616 |
| | | A "Fresnel integral" on classical Wiener space | 616 |
| | | The Banach algebras S and $\mathcal{F}(\mathcal{H}_1)$ are the same | 620 |
| | | Consequences of the close relationship between S and $\mathcal{F}(\mathcal{H}_1)$ | 621 |
| | | More functions in $\mathcal{F}(\mathcal{H}_1)$ | 622 |
| | | A unified theory of Fresnel integrals: Introductory remarks | 623 |
| | | Background material | 624 |
| | | A unified theory of Fresnel integrals (continued) | 626 |
| | | The Fresnel classes along with quadratic forms | 629 |
| | | The classes $\mathcal{G}^q(\mathcal{H})$ and $\mathcal{G}^q(B)$ | 630 |
| | | Quadratic forms extended | 631 |
| | | Functions in the Fresnel class of an abstract Wiener space: Examples of | |
| | | abstract Wiener spaces | 632 |
| | | Fourier-Feynman transforms, convolution, and the first variation for | |
| | | functions in S | 636 |
| | | B. References to further approaches to the Feynman integral | 636 |

xviii CONTENTS

| 20.2 | The influence of heuristic Feynman integrals on contemporary | |
|-------------|--|-----|
| | mathematics and physics: Some examples | 637 |
| | The heuristic Feynman path integral | 638 |
| | A. Knot invariants and low-dimensional topology | 639 |
| | The Jones polynomial invariant for knots and links | 639 |
| | Witten's topological invariants via Feynman path integrals | 641 |
| | Further developments: Vassiliev invariants and the | |
| | Kontsevich integral | 654 |
| | B. Further comments and references on subjects related to the | |
| | Feynman integral | 659 |
| | Supersymmetric Feynman path integrals and the Atiyah-Singer inde | х |
| | theorem | 659 |
| | Deformation quantization: Star products and perturbation series | 674 |
| | Gauge field theory and Feynman path integrals | 682 |
| | String theory, Feynman–Polyakov integrals, and dualities | 688 |
| | What lies ahead? Towards a geometrization of Feynman path | |
| | integrals? | 695 |
| References | | 697 |
| Index of sy | mbols | 745 |
| Author ind | ex | 750 |
| Subject ind | ex | 756 |

INTRODUCTION

The main purpose of this book is to provide a mathematical treatment of the Feynman path integral and the related subject of Feynman's operational calculus for noncommuting operators. The former subject is more widely known than the latter and has the reputation of being a formidable and rather elusive mathematical topic.

We will keep this introductory chapter, especially Section 1.1, nontechnical and relatively brief as far as possible. A detailed table of contents is provided and additional introductory chapters are included in the book in appropriate places. The main two are Chapters 7 and 14, dealing, respectively, with the first and second subjects:

Chapter 7, entitled "The Feynman integral: Heuristic ideas and mathematical difficulties", provides an introduction to quantum theory mainly from the perspective of the physicist Richard Feynman. Further, it points out why the Feynman "integral" is a difficult subject and shows how Feynman's ideas have led to the mathematical approaches to the Feynman integral which are used in Chapters 11–13 and 15–18.

Chapter 14 provides an introduction to Feynman's operational calculus for noncommuting operators, the subject of Chapters 15–19, and indicates how the Feynman integral and Feynman's operational calculus are related both in the present theory and in their historical development.

1.1 General introductory comments

Feynman's path integral

I find Feynman's formula to be very beautiful. It connects the quantum mechanical propagator, which is a twentieth-century concept, with the classical mechanics of Newton and Lagrange in a uniquely compelling way.

Mark Kac, 1984 [Kac5, p. 116]

Bohr got up and said: "Already in 1925, 1926, we knew that the classical idea of a trajectory or a path is not legitimate in quantum mechanics; one could not talk about the trajectory of an electron in the atom, because it was something not observable." In other words, he was telling me about the uncertainty principle. It became clear to me that there was no communication between what I was trying to say and [what] they were thinking. Bohr thought that I didn't know the uncertainty principle, and was actually not doing quantum mechanics right either. He didn't understand at all what I was saying. I got a terrible feeling of resignation. I said to myself, "I'll just have to write it all down and publish it, so that they can read it and study it, because I know it's right! That's all there is to it.

Richard P. Feynman, reminiscing about the 1948 Pocono conference.

(Quoted in [Me, p. 248].)

We begin with Feynman's famous heuristic formula [Fey1,2] for the evolution of a nonrelativistic quantum system:

$$\frac{1}{K} \int_{C_{u,v}^{0,t}} \exp\left\{\frac{i}{\hbar} S(x)\right\} \mathcal{D}x, \qquad (1.1.1)$$

where $i = \sqrt{-1}$. We will make some comments about this formula here, but a much more thorough discussion will be given in Chapter 7.

In (1.1.1), $C_{u,v}^{0,t}$ is the space of all real-valued (more generally, \mathbb{R}^d -valued) continuous functions x on [0, t] such that x(0) = u and x(t) = v. Further, $\mathcal{D}x$ represents a measure on $C_{u,v}^{0,t}$ which weighs all paths x equally (in much the same way as Lebesgue measure weighs all points in \mathbb{R} equally), \hbar is Planck's constant divided by 2π , and S(x) is the action integral associated with the path x; that is,

$$S(x) = \int_0^t \left\{ \frac{m}{2} \left[\frac{dx}{ds} \right]^2 - V(x(s)) \right\} ds. \tag{1.1.2}$$

The integrand in (1.1.2) is the *Lagrangian*; it equals, for each s in the time interval [0, t], the kinetic energy minus the potential energy at the point x(s).

Note that the potential V in (1.1.2) is real-valued, so that the integrand in (1.1.1) has a constant absolute value of one. Hence, it is the net interference effect as x ranges over the space of paths that determines the value of the oscillatory integral.

Feynman's ideas on the path integral (or "sum over histories") were ingenious and have had far-reaching consequences in many parts of physics, and more recently, of mathematics as well. At first, however, they seemed "crazy" to many physicists, including some famous ones (see [Me, §2.4]). Paths—and concepts that depend on paths, such as the Lagrangian and the action integral—play a crucial role in Feynman's formulation, whereas they had been "banned" (in light of the Heisenberg uncertainty principle) from the standard Hamiltonian approach to quantum dynamics (see Chapter 6).

The formula (1.1.1) seems hopeless at first to most mathematicians who come in contact with it. The "integral" in (1.1.1) is over a space of functions x "most" of which are nowhere differentiable, and yet the formula for the action S(x) in (1.1.2) involves calculating the derivative of x. Further, there is a mathematical theorem which implies that there is no countably additive measure on $C_{u,v}^{0,t}$ which weighs all paths equally. (See Section 3.1 for a closely related result.)

We should add that Feynman had some awareness of the mathematical difficulties just described above and concentrated throughout much of [Fey2] on a second approach (see Section 7.4) that begins with a discretization of the time interval [0, t]. (It enabled him, in particular, to replace the normalization constant K—which is ill-defined and for all practical purposes, infinite—by a suitable sequence of finite normalization constants.) This alternative approach involves fewer but still substantial mathematical difficulties.

The path integral of Feynman is not a Lebesgue integral; indeed, there is no "Feynman measure" (see Section 4.6, especially Theorem 4.6.1). At least for functions of physical interest, conditional convergence—instead of absolute convergence (as in the Lebesgue

theory)—is at the heart of the matter. Additionally, since the domain of integration of this oscillatory "integral" is a set of paths, the subject is intrinsically infinite dimensional. (Physically, the *cancellation effects* caused by the oscillatory nature of the Feynman integral correspond to interference effects between quantum-mechanical matter waves.)

The Feynman integral has been approached from many different points of view by mathematicians and physicists with varied background and interests. The resulting diversity has led to many different definitions of "the" Feynman integral. In this book, we address several (certainly not all) of these approaches in a setting appropriate for nonrelativistic quantum mechanics. In each of the cases considered, the existence of the Feynman integral is established under very general assumptions. The different approaches have their own domain of validity as well as their own strengths and weaknesses, as will be discussed further on in the book, especially in Chapters 11 and 13. However, under more restrictive but still quite general hypotheses, we will show that there is far more agreement than seems to have been previously realized between three of these approaches to the Feynman integral and the standard Hamiltonian approach to quantum dynamics. (See Section 13.4.)

Results on the Feynman integral for highly singular potentials are given in Chapters 11–13. Chapter 7, which was mentioned earlier, is crucial to an understanding of the Feynman integral. There, the physical background for nonrelativistic quantum mechanics is discussed from Feynman's point of view along with the way in which his ideas on the subject have led to several of the definitions of the Feynman integral which are used in Chapters 11–13. (Chapter 6 provides an extremely brief discussion of a few of the ideas which are common both to the usual Hamiltonian approach to quantum dynamics and to Feynman's approach.)

We close this part of the general introductory comments by providing more specific information on some issues that are central to the subject matter of this book through Chapter 13.

The following are shortcomings of many of the mathematical theories of the Feynman integral which are often pointed out:

- (1) The existence theory is not sufficiently general. In particular, many of the standard real-valued, time independent potentials $(V: \mathbb{R}^d \to \mathbb{R})$ which are used in modeling quantum systems are singular (for example, the attractive Coulomb potential) and do not fit within the theory.
- (2) Not much information is given about how the various approaches to "the" Feynman integral are related to one another or to the unitary group which gives the evolution of the quantum systems in the standard approach to quantum dynamics.
- (3) There is a shortage of satisfactory limiting theorems. Indeed, in some cases, no such theorems are available, while in others, the results do not seem natural from a physical point of view.

One of the strong points of the work here is that we give quite satisfactory responses to all three of these objections, especially for three of the four approaches to the Feynman integral which are developed in detail in this book. The Feynman integral defined via the Trotter product formula is shown to exist under very general conditions in

Corollary 11.2.22. Both the modified Feynman integral and the analytic-in-time operator-valued Feynman integral are shown to exist under even more general conditions in Corollary 11.4.5 and Theorem 13.3.1, respectively. Further, under the common conditions for their existence in the corollary and theorem just referred to, the modified Feynman integral and the analytic-in-time operator-valued Feynman integral not only exist but agree with each other and with the unitary group, as is shown in Corollary 13.4.1. Under the somewhat more restrictive conditions of Corollary 11.2.22, we will see in Corollary 13.4.2 that the Feynman integral via the Trotter Product Formula can be added to the list so that all three of these Feynman integrals exist and agree with one another and with the unitary group associated with the usual Hamiltonian approach to quantum dynamics.

Our limiting theorems for the three approaches to the Feynman integral referred to above are "dominated-type" convergence theorems. Since cancellation effects are intrinsic to the Feynman integral, there cannot be dominated convergence theorems in this subject that exactly parallel the Lebesgue dominated convergence theorem. However, in the most frequently used models in nonrelativistic quantum mechanics, it is only the potential energy function that may vary and our assumptions are that the sequence of functions $\{V_m\}$ is pointwise convergent (Lebesgue almost everywhere) and "dominated" in an appropriate sense (see (11.5.20) and (11.5.21) for example). The result for the modified Feynman integral, Theorem 11.5.19 [La12], is the key. The corresponding result for the analytic in time operator-valued Feynman integral, Corollary 13.4.3, is an easy corollary of Theorem 11.5.19 and Corollary 13.4.1. The convergence result for the Feynman integral via the Trotter product formula, Corollary 13.4.6, rests on Theorem 11.5.19 and Corollary 13.4.2 but also on some further considerations.

Although it is not especially difficult, Section 13.4 is quite pleasing because it brings together all of the positive results associated with items (1)–(3). (Note that we have omitted from the present discussion the analytic-in-mass operator-valued Feynman integral as studied in Sections 13.5 and 13.6. This material is interesting in its own right, but it is not readily compared with the three approaches above.)

The questions raised in (1)–(3) above are clearly central to the mathematical theory of the Feynman integral, but the answers provided in this book are not the only possible ones. Moreover, there are other important issues besides those implicit in (1)–(3). For example, the method of stationary phase is one of the heuristically appealing features of the Feynman path integral (see Chapter 7) but is not discussed rigorously anywhere in this book. However, it has been justified in the context of the Fresnel integral approach to the Feynman integral (see, for example, [AlHo2, Rez, AlBr1]).

Feynman's operational calculus

We turn now to the second topic in the title of this book, Feynman's operational calculus for noncommuting operators. A fuller introduction to this topic is given in Chapter 14.

It is easy to form functions of operators if the operators commute with one another. However, the subject becomes far more difficult when the operators fail to commute. Motivated by problems arising in quantum mechanics and quantum electrodynamics, Feynman ([Fey8], 1951) gave heuristic "rules" for forming functions of noncommuting operators. One of these "rules" says to treat the operators as though they commuted, once

a suitable time-ordering convention has been adopted. For example, Feynman writes such "equalities" as

$$\exp(A + B) = \exp(A) \exp(B), \tag{1.1.3}$$

even when A and B fail to commute.

The process of appropriately restoring the conventional ordering of the operators after the use of "equalities" such as in (1.1.3) above is referred to as "disentangling". This "disentangling process" is central to Feynman's operational calculus.

Feynman's "rules", as strange as they may seem, have led to useful results, notably the time-ordered perturbation series (or Dyson series) of quantum theory.

Feynman's work on his operational calculus is far from mathematically rigorous, as he himself noted. One of the challenges to mathematicians is to suitably interpret Feynman's ideas and to put them on a firm mathematical basis. Our work in Chapters 15–18 and in Chapter 19, respectively, discusses two ways of carrying this out and also further develops the subject in several directions.

What led Feynman to his operational calculus? He wanted a path "integral" in order to calculate perturbation series in quantum electrodynamics, but he had no such integral in that setting. His operational calculus was motivated initially by a desire to find methods of calculation which would generalize those which could be carried out in nonrelativistic quantum mechanics via his path "integral".

The operational calculus for noncommuting operators which Feynman discovered generalizes some aspects of path integration. This suggests that in settings where mathematically rigorous path integrals are available, it might be possible to use such integrals to interpret and make rigorous Feynman's operational calculus. Indeed, this is what we do in Chapters 15–18 using the Wiener and Feynman path integrals.

Feynman's operational calculus via the Feynman and Wiener integrals

Feynman's operational calculus, the Feynman integral and the Wiener integral all come together in Chapters 15–18 as well as in Sections 14.3–14.5. Chapters 15, 16 and 18 are based on joint work of the authors; much of this material can be found in [JoLa1] and [JoLa4], respectively. Chapter 17 is adapted from the following papers of the second author [La15, La18, La16].

The Wiener process (or Brownian motion) does not appear in the title of this book, but it—along with the associated Wiener measure and integral—appears repeatedly in this work. It plays an especially important role in Chapters 7 and 12–18. Chapters 3 and 4 present the information that we will need about Wiener measure from an analyst's point of view. A short Chapter 2 discusses physical Brownian motion and relates it to its mathematical model, the Wiener process. In Chapter 5, another short chapter, we give a very brief discussion of a more probabilistic approach to the Wiener process.

The main emphasis in Chapters 15–18 is on using the Feynman and Wiener integrals to study Feynman's operational calculus in the quantum-mechanical and diffusion (alternatively, heat or probabilistic) settings, respectively. However, many of the results in Chapters 15–18 have an interest of their own as contributions to the Feynman and Wiener integrals, apart from their connection with Feynman's operational calculus.

We will now describe more precisely than above our approach to the operational calculus in this context. A more detailed overview of Chapters 15–18 is provided in Chapter 14, especially in Sections 14.3–14.5.

The functions on the space of continuous paths on [0, t] that are Wiener and Feynman integrated in Chapters 15–18 belong, for each time t > 0, to the "disentangling algebra" A_t . This commutative Banach algebra consists of certain infinite sums of finite products of functions of the form

$$F(x) = F_{\theta,\eta}(x) := \int_{[0,t]} \theta(s, x(s)) \eta(ds), \tag{1.1.4}$$

where θ (often thought of as a time-dependent potential) is a complex-valued function on $[0, t) \times \mathbb{R}^d$ and η is a bounded Borel measure on [0, t). The function $\exp(F)$ is an important example of a function in \mathcal{A}_t . (It is called the "Feynman-Kac functional with Lebesgue-Stieltjes measure" η ; see Chapter 17. More generally, the elements of \mathcal{A}_t will often be referred to as "Wiener functionals" in Chapters 14-18.)

The operator-valued path integral of $F \in \mathcal{A}_t$ is denoted $K_{\lambda}^t(F)$. For $\lambda > 0$ (the diffusion case), $K_{\lambda}^t(F)$ is defined as a Wiener integral and then extended first via analytic continuation in λ to \mathbb{C}_+ , the open right half-plane, and then via continuity to $\mathbb{C}_+^{\sim} := \overline{\mathbb{C}_+} \setminus \{0\}$. When λ is purely imaginary (the quantum-mechanical case), $K_{\lambda}^t(F)$ is the "Feynman integral" of F. [This is the analytic (in mass) operator-valued Feynman integral of F; see Definition 15.2.1 for a more precise statement.]

The disentangling process is carried out in Chapters 15–18 by calculating the path integral $K_{\lambda}^{I}(F)$ for $\lambda > 0$ and then extending the result to $\lambda \in \mathbb{C}_{+}^{\infty}$. One need not invoke Feynman's "rules" explicitly in this setting; the necessary time-ordering is done naturally (but not always easily) while calculating the functional integrals.

The disentangled operators $K_{\lambda}^{I}(F)$ are expressed as time-ordered perturbation expansions or "generalized Dyson series". Generalized Feynman diagrams (see Section 15.6) provide a visual aid for keeping track of the terms of a generalized Dyson series. (These diagrams can be complicated in their own right but they generalize the simple diagrams of nonrelativistic quantum mechanics and not those of quantum electrodynamics.)

The work in Chapters 15–18 (and also in Chapter 19) not only interprets Feynman's ideas and makes them rigorous, but also extends them in several different ways. Non-commutative operations * and \dotplus on the family of disentangling algebras $\{\mathcal{A}_t\}_{t>0}$ are introduced in Chapter 18. They can be thought of as a noncommutative multiplication and addition, respectively, on the space of Wiener functionals; see Section 18.3. Such operations—introduced by the authors in [JoLa3,4]—were not envisioned by Feynman but they fit nicely into the operational calculus in various ways. If $F \in \mathcal{A}_{t_1}$ and $G \in \mathcal{A}_{t_2}$, then we know that the operators $K_{\lambda}^{t_1}(F)$ and $K_{\lambda}^{t_2}(G)$ can be disentangled via generalized Dyson series. It is natural to ask if the product of $K_{\lambda}^{t_1}(F)$ and $K_{\lambda}^{t_2}(G)$ can also be disentangled. It can; in fact (Theorem 18.5.6 and Corollary 18.5.7), $F * G \in \mathcal{A}_{t_1+t_2}$ and for all $\lambda \in \mathbb{C}_{+}^{\infty}$.

$$K_{\lambda}^{t_1+t_2}(F*G) = K_{\lambda}^{t_1}(F)K_{\lambda}^{t_2}(G). \tag{1.1.5}$$

Since we can show that

$$\exp(F \dotplus G) = \exp(F) * \exp(G) \tag{1.1.6}$$

on the level of the functionals, we immediately deduce from (1.1.5) that, on the level of the operators.

$$K_{\lambda}^{t_1+t_2}(\exp(F \dotplus G)) = K_{\lambda}^{t_1}(\exp(F))K_{\lambda}^{t_2}(\exp(G)).$$
 (1.1.7)

Note that (1.1.6) formally resembles Feynman's paradoxical formula (1.1.3) but involves the noncommutative operations * and \dotplus on the disentangling algebras.

The family of commutative disentangling algebras $\{A_t\}_{t>0}$ —equipped with the non-commutative operations * and \dotplus along with the (operator-valued, analytic-in-mass) Feynman integrals $K_{\lambda}^{t}(\cdot)$ —forms a rich interlocking algebraic and analytic structure that enables us to explore more deeply the noncommutative aspects of Feynman's operational calculus.

Our systematic use of measures as in (1.1.4) contributes significantly to the richness of Feynman's operational calculus. Different measures can provide different directions for disentangling. For example, what is one exponential function of a sum of commuting operators becomes infinitely many different exponential functions of a sum of noncommuting operators. This leads in Chapter 17, entitled "The Feynman-Kac formula with a Lebesgue-Stieltjes measure and Feynman's operational calculus" and based on work of Lapidus in [La14-18], to the solution of a wide variety of evolution equations which can incorporate both discrete and continuous phenomena.

Feynman's operational calculus and evolution equations

Another approach to Feynman's operational calculus is considered in Chapter 19, based on joint work of the authors with Brian DeFacio ([dFJoLa1] and especially [dFJoLa2]). The setting is much more general than in Chapters 15–18, but, on the other hand, attention is focused almost exclusively on exponentials of sums of noncommuting operators. In [Fey8] and in the papers which led up to it, the emphasis was also on such exponential functions. This particular focus came from Feynman's desire to calculate formulas for the evolution of physical systems.

The operators that appeared as the arguments of the exponential function in Feynman's work were associated with the different forces involved in the physical problem. Feynman seemed to have complete confidence that applying his "rules" to such exponential expressions would yield a formula for the evolution of the physical system at hand. The main results of Chapter 19. Theorems 19.5.1 and 19.6.1, justify (in a mathematical sense) Feynman's confidence (under a certain rather general set of hypotheses) by showing that the disentangled exponential expression gives the unique solution to the associated evolution equation. Our method is to use Feynman's heuristic ideas to "disentangle" the exponential expression: we then prove that the disentangled expression makes sense and satisfies the evolution equation.

We hope that the combination of some simple examples of disentangling found in Chapter 14, the more complicated calculations from Chapter 19 that were just referred to above, along with some additional examples that are provided in Section 19.7, will help to clarify Feynman's heuristic "rules" for the reader. Chapters 15–18 will also be helpful in this regard. Although the disentangling is carried out in these chapters in the process of calculating the Wiener and Feynman integrals, one can see clearly the connections with Feynman's time-ordering ideas both in the details of the calculations and in the resulting answers.

Further work on or related to the Feynman integral: Chapter 20

Chapter 20, our last chapter, has a very different character from the rest of this book. Our main focus in regard to the Feynman integral will be on operator-valued approaches. However, in Section 20.1, we will give a brief expository account (without proofs) of scalar-valued approaches to the Feynman integral which involve "transform assumptions". A great deal of work on the Feynman integral has been along these lines since the 1976 monograph of Albeverio and Hoegh-Krohn [AlHo1] on the "Fresnel integral".

In Section 20.2, our main concern is with the connections between the "heuristic Feynman integral" and a variety of further topics in contemporary mathematics and physics. The greatest emphasis will be on Section 20.2.A where we discuss Witten's heuristic Feynman integral [Wit14] and its influence on the subjects of knot theory and low-dimensional topology. In Section 20.2.B, we briefly discuss the relationship between heuristic path integrals and four additional topics: The Atiyah–Singer index theorem, deformation quantization, gauge field theory, and string theory. We should stress that the mathematical existence of the "Feynman integrals" used in Section 20.2 has usually not been established. We should also caution the reader that the authors are far from being experts on the subjects involved in Section 20.2.

Given its special nature, Chapter 20 will be excluded from our discussion in the remainder of this introduction.

Section 1.1, with the exception of its last subsection, has been a brief introduction to the main topics of this book. Next we turn to a discussion of some of the themes that are repeated in several places in this work. An ordered (rather than thematic) and quite detailed list of the topics treated in this book can be found in the list of contents; the latter has been written partly with this goal in mind. Section 1.2 below is somewhat more technical than Section 1.1. Depending on their background, some readers may wish initially to go over parts of this material quickly and then return to it at a later time.

1.2 Recurring themes and their connections with the Feynman integral and Feynman's operational calculus

There are a number of subjects related to those in the title of this book which will play an important role and will reappear frequently; the Wiener process has already been mentioned in this connection. Product formulas, such as the Trotter Product Formula and the product formula for imaginary resolvents discussed in detail in Chapter 11, certainly fall into this category as well.

Product formulas and applications to the Feynman integral

Perhaps the approach to the Feynman integral which is most straightforwardly motivated by Feynman's original paper ([Fey2], 1948), is the approach using the Trotter product

formula. It is Trotter's formula for the case of unitary groups that is used. Ignoring some technicalities, this result says that if A and B are (unbounded, noncommuting) self-adjoint operators on a Hilbert space \mathcal{H} and if A + B is essentially self-adjoint (i.e., if it has a unique self-adjoint extension), then

$$e^{-it(\overline{A+B})} = \lim_{n \to \infty} \left(e^{-i\frac{t}{n}A} e^{-i\frac{t}{n}B} \right)^n, \tag{1.2.1}$$

where here, by the operator $\overline{A+B}$ on the left-hand side of (1.2.1), we mean the unique self-adjoint extension of the algebraic sum A+B.

When (1.2.1) is applied to the Feynman integral, the Hilbert space \mathcal{H} will be $L^2(\mathbb{R}^d)$, and we will take, after normalizing the physical constants, $A=-\frac{1}{2}\Delta=H_0$ (the free Hamiltonian), where Δ denotes the Laplacian on \mathbb{R}^d . Further, we will let B=V, the operator of multiplication by the potential energy function. (The "potential" $V:\mathbb{R}^d\to\mathbb{R}$ is a suitable real-valued function on \mathbb{R}^d .) Finally, we let $H=\overline{A+B}=\overline{H_0+V}$ denote the Hamiltonian or energy operator associated with V. Then, when applied to an appropriate wave function φ , the left-hand side of (1.2.1), namely, $\psi(t,\cdot):=e^{-itH}\varphi$, yields the unique solution of the Schrödinger equation

$$i\frac{\partial\psi}{\partial t} = -\frac{1}{2}\Delta\psi + V(\xi)\psi = H\psi \ (\xi \in \mathbb{R}^d, t \in \mathbb{R}). \tag{1.2.2}$$

with initial state $\psi(0,\cdot) = \varphi$ in the domain of H.

The approach to the Feynman integral via the Trotter product formula is the first of two approaches which appeared in Nelson's paper ([Ne1], 1964). An informal explanation of the connection between [Fey2] and the Trotter product formula is given in Sections 7.2 and 7.5 and a precise discussion with proofs of a result which is more general than the one in [Ne1] appears in Sections 11.1 and 11.2.

Inspired by the product formula of Trotter and the work of Nelson, Lapidus found [La11] a "product formula for imaginary resolvents" and used it to define and establish the existence of the "modified Feynman integral". The result of Lapidus goes well beyond the case where A + B is essentially self-adjoint; in fact, his formula involves the "form sum" A + B of the operators A and B. Also, the unitary operators $e^{-i\frac{t}{n}A}$ and $e^{-i\frac{t}{n}B}$ on the right-hand side of (1.2.1) are replaced by the imaginary resolvents $[I + i(t/n)A]^{-1}$ and $[I + i(t/n)B]^{-1}$, respectively. Thus, we have the "product formula for imaginary resolvents" (see Section 11.3, especially Theorem 11.3.1 and Corollary 11.3.7):

$$e^{-it(A+B)} = \lim_{n \to \infty} ([I+i(t/n)A]^{-1}[I+i(t/n)B]^{-1})^n.$$
 (1.2.3)

(If A+B is essentially self-adjoint, as in the hypothesis of the product formula for unitary groups (1.2.1), then the form sum A + B coincides with $\overline{A+B}$, the unique self-adjoint extension of the algebraic sum A+B—and so the left-hand side of (1.2.3) coincides with that of (1.2.1); see Proposition 11.2.10(ii).)

When the product formula (1.2.3) is applied to define and establish the convergence of the modified Feynman integral, we obtain much as before a solution to the Schrödinger equation, but now with the Hamiltonian given by the form sum of H_0 and V; i.e., $H = H_0 \dotplus V$. (See Section 11.4, including Definition 11.4.4.)

In the setting we have been considering, the potential is a real-valued and time-independent function V and the Hamiltonian is obtained by "adding" V to H_0 , with the sum allowed to be a form sum. The maximum domain of validity for V in this framework is (as we will see in Section 11.4) exactly the same for the modified Feynman integral as it is in the Hamiltonian approach to quantum dynamics. Further, this maximum domain of validity is physically natural; the "form domain" of the Hamiltonian $H = H_0 \dotplus V$ consists precisely of those functions $\varphi \in L^2(\mathbb{R}^d)$ which have finite total (i.e., kinetic + potential) energy. Looking ahead and considering the same setting, the maximal domain of validity for V in the case of the analytic-in-time operator-valued Feynman integral agrees with the other two. It should be added, however, that the modified Feynman integral extends nicely to the case of complex potentials V (see Section 11.6) whereas a corresponding theorem has not been proved (and may not be true) for the analytic-in-time operator-valued Feynman integral considered in Sections 13.3 and 13.4.

An advantage of the generality of the modified Feynman integral is that it allowed Lapidus to obtain in [La12] a very general stability theorem (relative to the potential) and to deduce a "dominated-type convergence theorem" appropriate for this context. (See Section 11.5.)

The results leading to the definition of the "Feynman integral via TPF" [Ne1] are discussed in Section 11.2, while those concerning the "modified Feynman integral" [La1-2, La6-13] and various extensions of its definition (notably, to C-valued potentials [BivLa]) are presented in Sections 11.3-11.6. In addition, we mention that Sections 13.5 and 13.6, respectively, describe analytic (in mass) versions of these two approaches to the Feynman integral. Product formulas of various types—not themselves consequences of (1.2.1) or of (1.2.3)—also play a prominent role in these sections.

Feynman-Kac formula: Analytic continuation in time and mass

Mark Kac heard Feynman speak about his path integral in 1947. Kac realized that if time t in Feynman's formula is replaced by -it ("imaginary time" from the perspective of quantum physics), then the expression involved before the limit is taken is equal to a Wiener integral, a true integral in the Lebesgue sense with respect to the countably additive Wiener measure m. The powerful results of the Lebesgue theory of integration can then be used to rigorously justify the calculation of the limit. One outcome of all this is the famous Feynman-Kac formula. (A detailed proof of a very general version of the result is given in Chapter 12, based on work of B. Simon in [Si9].)

Kac's discovery expresses the solution to the heat equation as a certain Wiener integral. More precisely, if the "Feynman-Kac functional" F is given by

$$F(x) := \exp\left\{-\int_0^t V(x(s))ds\right\},\tag{1.2.4}$$

then for $t \geq 0$ and $\xi \in \mathbb{R}^d$, we have

$$u(t,\xi) := (e^{-tH}\varphi)(\xi)$$

$$= \int_{C_0^t} F(x+\xi)\varphi(x(t)+\xi)d\mathfrak{m}(x), \qquad (1.2.5)$$

where m denotes Wiener measure on the space C_0^t of continuous paths x such that x(0) = 0. The left-hand side of (1.2.5) yields the unique solution, $u(t, \cdot) = e^{-tH}\varphi$, at time $t \ge 0$ of the heat (or diffusion) equation

$$-\frac{\partial u}{\partial t} = -\frac{1}{2}\Delta u + V(\xi)u = Hu \ (\xi \in \mathbb{R}^d, t \ge 0), \tag{1.2.6}$$

with initial condition $u(0, \cdot) = \varphi$. Here, as before, $H = H_0 + V$, with $H_0 = -\frac{1}{2}\Delta$. Note, however, that we now use the heat semigroup e^{-tH} to represent the solution to the heat equation (1.2.6) whereas we have used earlier the unitary group e^{-itH} to represent the solution to the Schrödinger equation (1.2.2).

The "Feynman-Kac formula" (1.2.5) has been extremely useful for a variety of purposes, both in mathematics and in physics (see Section 7.6 for a brief discussion of this along with some references), but it does not by itself resolve the problem of making sense of the Feynman integral since the change from t to -it takes us from quantum theory and the Schrödinger equation to the heat equation. The Feynman-Kac formula does, however, suggest an approach to the Feynman integral. Start with imaginary time and the theoretically powerful Wiener integral and define the Feynman integral by analytically continuing to real time. Indeed, operator-valued analytic continuation in time is another of the approaches to the Feynman integral which will be discussed in detail in this book. These results on the analytic-in-time Feynman integral (at the level of generality found here) are due to Johnson [Jo6] and are the subject of Sections 13.2 and 13.3. We should mention that what is imaginary time from the point of view of quantum theory is real time from the perspective of the heat equation. We shall adopt the latter point of view in Chapter 13 (Sections 13.2, 13.3 and 13.7) and analytically continue from real time to purely imaginary time—going in the process from the Wiener integral to the Feynman integral.

We remark that Section 13.7 gives a brief discussion of an extension (see [AlJoMa]) of the analytic-in-time operator-valued Feynman integral which is based on the theory of "additive functionals of Brownian motion" (see [Fuk, FukOT]) and Feynman-Kac formulas in which, for example, the potential V can be replaced by a suitable measure on \mathbb{R}^d .

The last of the approaches to the Feynman integral which will be treated in detail in this book is operator-valued analytic continuation in mass. Again, one starts with the Wiener integral but this time, the analytic continuation is in a mass parameter (or alternately, in a variance parameter). The connection between Feynman's ideas and the approach to the Feynman integral via operator-valued analytic continuation in mass is discussed in an informal way in Section 7.6, with the approach via the Trotter product formula serving to link the two.

The precise discussion of the analytic-in-mass operator-valued Feynman integral is given in Section 13.5. The crucial starting point for this work is Nelson's second approach developed in [Ne1]. An earlier paper by Cameron ([Ca1], 1960) used scalar-valued analytic continuation in mass; the key contribution of [Ca1] was the proof that there is no countably additive "Wiener measure" with a complex variance parameter (see Theorem 4.6.1). This result corrected an error in [GelYag], an interesting and even earlier paper which used analytic continuation.

Various extensions of Nelson's results are given in Sections 13.5 and 13.6. Among them, the reader will find hybrids which combine a suitable product formula with analytic continuation in mass. A comparison of the resulting analytic in mass Feynman integrals within their common domain of validity is provided towards the end of Section 13.6.

We remind the reader that the analytic-in-mass operator-valued Feynman integral will also be used in Chapters 15-18. Unlike the approaches in Chapter 13 via analytic continuation in mass, this Feynman integral exists for *every* (rather than Lebesgue almost every) value of the mass parameter. The class of functionals treated in Chapters 15-18 is, *in some respects*, much larger than in Chapter 13. However, in Chapters 15-18, no attempt is made to deal with potential functions with strong spatial singularities.

There are four different versions of the analytic-in-mass Feynman integral discussed in this book, as was alluded to above; in addition, three other approaches to the Feynman integral have already been discussed in this introduction. In the next two paragraphs, we indicate briefly what these are and where they are to be found.

The approaches to the Feynman integral that are discussed at any length in this book are all operator-valued. (Recall that we are not taking Chapter 20 into account in our present discussion.) Two of the analytic-in-mass approaches start from the Wiener integral when the mass parameter is real. One of these is discussed in the first part of Section 13.5; the other, which has quite different features, is defined in Section 15.2 and used throughout Chapters 15–18. The last two begin with product formulas for semigroups (in Section 13.5) and resolvents (in Section 13.6) to yield analytic-in-mass versions of the Feynman integral via TPF ([Kat7, BivPi]) and of the modified Feynman integral [BivLa], respectively.

The Feynman integral defined via the Trotter product formula for unitary groups is discussed in Section 11.2 and the modified Feynman integral (defined via a product formula for imaginary resolvents established in Section 11.3) is treated in Sections 11.4–11.6. Finally, the analytic-in-time Feynman integral appears in Sections 13.2 and 13.3, with an extension given in Section 13.7.

The role of operator theory

As mentioned above, the approaches to the Feynman integral that will be discussed in detail in this book are all operator-valued. Further, there is always at least one unbounded operator involved; much of the time, it is $H_0 = -\frac{1}{2}\Delta$, the free Hamiltonian, although various physically meaningful substitutes for H_0 are allowed in Sections 11.4, 11.6, and Sections 13.5–13.6, and more abstract generators are considered in Chapters 11 and 19. In Sections 11.2–11.5, Chapter 12, Sections 13.2–13.4 and 13.7, the theory of (not necessarily bounded) self-adjoint operators and functions of them is sufficient for

our needs. These needs include various forms of the spectral theorem for unbounded self-adjoint operators as well as basic results about unbounded quadratic forms and form sums of operators. This background material is provided in Chapter 10 which is titled "Unbounded self-adjoint operators and quadratic forms". (See also Section 9.6 for introductory material on unbounded self-adjoint operators and the associated semigroups.) The spectral theorem enables us to define the functions e^{-itH} (the unitary group) and, if the spectrum of the self-adjoint operator H is bounded from below, the (self-adjoint) semigroup e^{-tH} . For us, in most applications, H is the Hamiltonian (or energy operator), a suitable self-adjoint extension of $H_0 + V$, where V is the potential. (More specifically, in Section 11.2, H is the unique self-adjoint extension of $H_0 + V$, and, more generally, it is the form sum of H_0 and V in Sections 11.3–11.5, Chapter 12, Sections 13.2–13.3 and 13.7.)

Self-adjoint operators—and the associated unitary groups or self-adjoint semigroups—are not adequate for everything that we will do. Strongly continuous (or (C_0)) semigroups of operators will be discussed in Chapter 9 (and in the brief and informal chapter that precedes it). Such semigroups (not necessarily associated with self-adjoint operators) will be used in Sections 11.1, 11.6, 13.5, 13.6, parts of Chapter 14 and throughout Chapter 19. They will also frequently be present in Chapters 15–18 but will be used in a more straightforward way there.

Connections between the Feynman-Kac and Trotter product formulas

The Feynman-Kac and Trotter product formulas have already been discussed above. but there are additional places in the book where these related formulas or variations of them appear. The Trotter product formula is the main tool in the crucial first step of the proof of the Feynman-Kac formula in Chapter 12. A variation of the Feynman-Kac formula, the "Feynman-Kac formula with a Lebesgue-Stieltjes measure", is—along with its connection with Feynman's operational calculus—the topic of Chapter 17, which describes part of the work in [La14-18]. A related product integral, a relative of the Trotter product formula, is discussed in Section 17.6 [La18,16]. Example 16.2.7 (in conjunction with Example 15.5.5) looks at the relationship between the Trotter product formula and the Feynman-Kac formula from the point of view of weak (or vague) convergence of measures. This broad perspective is informative even though the results are far less general than those proved in Chapters 11 and 12. A version of the Feynman-Kac formula which substantially extends the one in Chapter 12 is discussed briefly in Section 13.7. There, for example, the potential energy function can be replaced by certain measures (in the space rather than in the time variable, as in Chapter 17) which are singular with respect to Lebesgue measure. Finally, a Feynman-Kac formula for certain complex potentials is contained in the work presented in Sections 13.5 and 13.6.

Evolution equations

A fundamental concept of quantum mechanics is a quantity called the propagator, and the standard way of finding it (in the non-relativistic case) is by solving the Schrödinger equation. Feynman found another way based on what became known as the Feynman path integral or "the sum over histories" . . .

Mark Kac, 1984 [Kac5, p. 116]

The evolution of physical systems concerns us throughout this book, so it is not surprising that the subject of evolution equations is another recurring theme. Our point of view (following Feynman) is not, however, the usual one. Typically, the evolution equation comes first and is regarded as the model for the physical system. One then looks for a method to solve the evolution equation and the solution gives the evolving state. Our deviation from this point of view is perhaps seen most clearly in Chapter 19. The idea there is: Given the forces involved in the problem, write down and then "disentangle" the exponential of a sum of integrals (from, say, 0 to t) of associated time-ordered operators (see (19.4.8)). The resulting time-ordered perturbation series (see (19.3.14)) should give the evolution of the physical system. Of course, it is of mathematical and physical interest to know if this series solves some related evolution equation. Theorem 19.5.1 shows that this is so under a quite general set of assumptions.

As remarked earlier in this introduction, the approach to quantum dynamics provided by "the" Feynman path integral differs in several ways from the standard Hamiltonian approach. The point we wish to make here is that the path integral itself should give the evolving state. No evolution equation is needed ahead of time. Of course, it is of interest to know conditions under which the evolving state given by the Feynman integral satisfies the Schrödinger equation or some variation of it.

The different specific approaches to the Feynman integral discussed in this book have differing relationships with the standard Hamiltonian approach to quantum dynamics. Our first comments along these lines pertain to Chapter 17. Recall that in Chapters 15-18, the potentials can be time-dependent and complex-valued but are not allowed to have strong singularities in the space variables. If we take the appropriate Wiener integral involving the usual Feynman–Kac functional $\exp\{F_{\theta,l}(x)\}\$, where $F_{\theta,l}$ is given by (1.1.4) and l is Lebesgue measure on the time interval (0, t), we obtain a function of time and space which describes the evolution of a distribution of heat. By analytically continuing in mass (and making an adjustment in the potential), we arrive at a function giving a quantum evolution. These time evolutions are solutions to the heat and Schrödinger equations, respectively. In Chapter 17, we replace the Feynman–Kac functional $\exp\{F_{\theta,l}\}$ by the Feynman-Kac functional $\exp\{F_{\theta,\eta}\}$ (where $F_{\theta,\eta}$ is given by (1.1.4) and η is a Lebesgue-Stieltjes measure) and follow the procedure above. We show first that the resulting evolutions involve an interesting variety of discrete and continuous phenomena and then also that they are solutions to correspondingly adjusted versions of heat and Schrödinger equations which are quite different from the usual ones (see especially Sections 17.2 and 17.6).

Even though Feynman's approach to quantum dynamics does not depend a priori on the usual one, the method of proof for three of the specific approaches discussed in this book, the Feynman integral via the Trotter product formula (Section 11.2), the modified Feynman integral (Sections 11.3 and 11.4), and the operator-valued analytic-in-time Feynman integral (Sections 13.3 and 13.7), not only depend heavily on operator-theoretic results but also on the existence of the unitary group as established in the standard Hamiltonian approach. [In the case of the modified Feynman integral with complex (rather than real) potential studied in Section 11.6 ([BivLa]), the Schrödinger

operator must be defined appropriately and the associated time evolution is dissipative but in general, not unitary.]

The situation is quite different for the analytic-in-mass operator-valued Feynman integral, whether you begin on the real line with a Wiener integral (Section 13.5) or with product formulas (Section 13.6 and the last part of Section 13.5). Although operator techniques are still heavily involved, they are not the ones based on self-adjointness that are used commonly in quantum mechanics. Moreover, knowledge of the existence of the unitary group from the usual approach to quantum dynamics is not needed in the proof. In fact, for *extremely singular potentials* (see Examples 13.6.13 and 13.6.18), the analytic-in-mass operator-valued Feynman integral exists but the Hamiltonian approach does not, at least not in an unambiguous way.

Functions of noncommuting operators

The formation of functions of noncommuting operators is a theme which is implicit in the title of this book and which is of direct concern to us throughout Chapters 14-19. Although it is less obvious, the same subject is also involved in Chapters 6-13. For example, if A and B are commuting self-adjoint operators, there is no need for the Trotter product formula (1.2.1); we simply have $e^{-it(A+B)} = e^{-itA}e^{-itB}$. The Trotter product formula has sometimes been referred to as the noncommutative exponential law. (In light of our later work, especially in Chapters 17 and 19, it would be more accurate to describe it as an especially important example but just one of many noncommutative exponential laws.) The spirit of the theory of semigroups of operators is that it is the theory of forming the "exponential function" of operators. In practice for us (and in general), the operator to be "exponentiated" is often of the form A + B, where A and B do not commute. The Feynman-Kac formula expresses the heat semigroup $e^{-tH} = e^{-t(H_0+V)}$ (where $H_0 = -\frac{1}{2}\Delta$ is the free Hamiltonian and V is the operator of multiplication by the potential V) as a certain Wiener integral. (See equations (1.2.5) and (1.2.4) above.) In some sense, this formula can be thought of as providing a way to handle the fact that the operators H_0 and V do not commute.

Time-ordered perturbation series

In [Fey8] and in the work in this book on Feynman's operational calculus, the disentangled functions of operators are more often than not expressed as time-ordered perturbation series. In Chapters 14–19, such series appear repeatedly. They are most often referred to as generalized Dyson series in Chapters 15–18. Indeed, special cases of the perturbation series in all of Chapters 14–19 coincide with the classical Dyson series of nonrelativistic quantum mechanics.

In Chapter 15 (and then throughout Chapters 16–18), our generalized Dyson series play a crucial role in *defining* the operators $K_{\lambda}^{t}(F)$ involved, especially in the quantum-mechanical (or Feynman) case where a bona fide path integral (such as the Wiener integral) is no longer available. In turn, these perturbation expansions—which can be thought of, in some extended sense, as providing a "sum over all possible histories" of a quantum particle—are very helpful mathematical tools and enable us to derive various properties with relative ease. (See, for example, Sections 15.3, 15.5, and Chapter 16.)

When F is an exponential functional (see Chapter 17), they also play a key role in deriving the evolution equation (either in differential or integral form) satisfied by $t \mapsto K_{\lambda}^{t}(F)$. (This is especially true in the quantum-mechanical case.)

[For simplicity, we will limit ourselves here to the setting of Chapters 15–18. We point out, however, that despite certain differences due to the generality of the assumptions made in Chapter 19 and the absence of any kind of path integral in that framework, our above comments regarding the *definition* of the operators involved and the derivation of a corresponding *evolution equation* remain valid in the setting of Chapter 19 as well.]

At this point, it may be helpful to recall from our discussion in Section 1.1 that in Chapter 15, given a Wiener functional F in the disentangling algebra A_t , we associate with it an operator $K_{\lambda}^{l}(F)$, called the analytic (in mass) Feynman integral of F, which can be disentangled via a generalized Dyson series. [Briefly, the bounded linear operator $K_{\lambda}^{l}(F)$ is defined as a genuine Wiener (path) integral in the diffusion case when λ is real, and then, for complex λ , by analytic continuation followed by passage to the limit along the imaginary axis of the resulting perturbation expansion.] Consequently, the time-ordered perturbation series for $K_{\lambda}^{l}(F)$ has the same general expression as a function of the parameter λ both in the diffusion (or probabilistic) case (λ real and positive) and in the quantum-mechanical (or Feynman) case (λ purely imaginary and nonzero). This fact enables us to deal with these two situations in parallel in much of Chapters 15–18. (Notable exceptions occur in Sections 16.2 and 17.6.)

There is a last general comment that we wish to make about the "disentangling" provided by our generalized Dyson series: It is not necessarily unique; indeed, a given operator $K_{\lambda}^{I}(F)$ can be represented in many different ways via a time-ordered perturbation series, some of which may be more suitable than others in a given situation. (See especially Section 15.5 for various examples; see also, for instance, Section 17.6.) We stress that in spite of this fact, the operator $K_{\lambda}^{I}(F)$ associated with a function F in the "disentangling algebra" A_{t} is always defined uniquely (and hence unambiguously). In some suitable sense, the mapping K_{λ}^{I} (defined in Section 15.7) can be thought of as a quantization map from the commutative disentangling algebra A_{t} to a noncommutative algebra of (bounded linear) operators. [See especially Chapter 18 (including Appendix 18.6), where the action of the noncommutative operations * and + on the family of disentangling algebras $\{A_{t}: t > 0\}$ is taken into account.]

The use of measures

Measures and their associated integrals enter into this book in various ways. We mention some of these here and emphasize those which are less widely familiar but will be especially important to us.

Two of the definitions of "the" Feynman integral that are stressed in this book start with the Wiener integral. Our purpose in Chapter 3 is to give the reader who is not acquainted with Wiener measure some idea of how it can be constructed and some familiarity with the properties of the Wiener process that will be needed in subsequent work. The construction follows the pattern of Lebesgue measure on the line, a topic familiar to most mathematicians. It begins with the definition of the measure of an "interval" and ends with an application of the Carathéodory extension theorem.

We expect that many potential readers will be familiar with the results of Chapter 3 and with Lévy's quadratic variation law which is the subject of Section 4.1. However, we anticipate a much lower degree of familiarity with most of the rest of Chapter 4 which deals with such topics as the family of scaled Wiener measures $\{m_{\sigma}: \sigma>0\}$, scale-invariant measurability [JoSk7] and the refined equivalence classes of functions that are needed for a careful discussion of the Feynman integral obtained via analytic continuation in mass. This definition of the Feynman integral will concern us in Section 13.5 (the second approach in [Ne1]) as well as throughout Chapters 15–18.

Measures on subintervals of \mathbb{R} (Lebesgue–Stieltjes measures) are used systematically throughout Chapters 14–19 in connection with Feynman's operational calculus. They serve not only to assign weights but also to time-order the integrands which are usually (perhaps after some preliminary steps) products of noncommuting operators. The measures give directions for "disentangling", and a different set of measures can yield very different results. The first few pages of Section 14.2 (through Example 14.2.1) can be read independently of all of the earlier material in this book and will provide the reader with a discussion of Feynman's heuristic "rules" and an extremely simple example of the points made above.

1.3 Relationship with the motivating physical theories: background and quantum-mechanical models

What does this book have to say about the physical theories which motivate it? The reader will not find here applications to concrete and detailed physical problems of the mathematical results contained within. However, in certain respects, we do discuss in a number of places related physical theories and especially quantum mechanics.

Physical background

A discussion of the relevant physical background is provided in key places. Most importantly, Feynman's way of looking at quantum mechanics and his path integral and how this has led to the approaches to the Feynman integral found in this book is the subject of Chapter 7. Chapter 6 contains an extremely brief discussion of some parts of the usual Hamiltonian approach to quantum dynamics; this chapter is included partly for the sake of contrast but also because the two approaches have, of course, some common features. It seems to us that it is difficult to get an appreciation for the mathematics of the Feynman integral without at least some understanding of the physical background.

As noted earlier, this book contains a good deal of information about the Wiener integral (see Chapters 2–5, 7 and 12–18). Much of this material, apart from Chapter 3, Section 4.1 and Chapter 5, is not the standard fare but consists of special topics related to the two items in the title of this book. Chapter 2 discusses the character of physical Brownian motion and the way in which that led Norbert Wiener, through the work of Brown. Einstein and Perrin. to what is now known as the Wiener process, the mathematical model of Brownian motion.

Chapter 14 is an introduction to Feynman's operational calculus. Some discussion of the physical problems that led Feynman to this calculus can be found there, but much less than one might guess. Why is that?

The primary purpose of the paper [Fey8], "An operator calculus having applications in quantum electrodynamics", was to present the ideas and rules which Feynman had developed in connection with [Fey5–7] for forming functions of noncommuting operators. While most of the examples in [Fey8] are from quantum theory, Feynman was well aware that he had found a computational technique with implications beyond that particular setting. [In fact, this point was stressed repeatedly by Richard Feynman himself in a number of conversations with the second-named author (M. L. L.), during the first of which (in about 1981) Feynman mentioned his paper [Fey8] on the subject and urged M. L. L. to develop his operational calculus and to put it on a firm mathematical basis.] Chapter 14 is an exposition of these *mathematical* (but not mathematically rigorous) ideas of Feynman and how they will be interpreted, extended and developed with mathematical rigor in Chapters 15–19.

[The reader may be aware of Feynman's sometimes negative comments about some of the mathematicians' musings (see, for example, [Fey16,17]). However, he/she may wish to contrast this impression with Feynman's comments in [Fey8, p. 108] regarding the need for mathematical rigor and for further mathematical exploration of his "operator calculus". (See the second quote from [Fey8] at the very beginning of Chapter 14, which is in complete agreement with the second author's conversations with Feynman.) Perhaps it is appropriate at this point to add two more personal recollections. When asked by a physics Ph.D. student how much mathematics he needed to learn, Feynman answered without hesitation: "As much as possible." (This was witnessed by the second author in Los Angeles in 1981.) Finally, and to give a more balanced view, when during his 1983 UCLA public lectures for a general scientifically curious audience (of which his book QED, [Fey15], is an edited version), he was asked what were the relationships between mathematicians and physicists, he began his answer (approximately) as follows: "They are very good friends, but they do not consider the same problems, and they do not have the same point of view. The mathematician looks at a very broad area and is interested in everything related to it. The physicist, on the other hand, who is interested in certain specific questions, can go much further in some particular directions..."]

The discussion of physical background and physical interpretation of results goes beyond the introductory chapters mentioned above. It can be found in various places throughout the book. We mention Chapters 11, 13, 15, 16 and especially, Chapters 17 and 19.

Highly singular potentials

A variety of quantum-mechanical models are discussed in this book. These include in Chapters 11 and 13 highly singular potentials V and the standard Hamiltonian

$$H = -\frac{1}{2}\Delta + V. \tag{1.3.1}$$

In (1.3.1), V denotes the operator of multiplication by a time independent, real-valued potential energy function V. [The precise form of H when the mass m and $\hbar = (Planck's constant)/2\pi$ are not normalized is given in (6.4.1). For the case of an N-particle system where the jth particle has mass m_j , $j = 1, \ldots, N$, see (6.4.2).] The inclusion of highly singular potentials in the approaches to the Feynman integral discussed in Chapters 11

and 13 is a major advantage of those approaches. Some of the most basic potentials of quantum mechanics such as the Coulomb potential are singular in the space variables. (See [FrLdSp] for a detailed account from a physicist's point of view of the role of singular potentials in quantum theory.)

A discussion of highly singular central potentials is provided in Example 11.4.7 and pursued in Example 13.6.13. The interesting special case of the inverse-square potential is treated in Example 13.6.18.

We give in Example 11.4.12 and in parts of Sections 13.5 and 13.6 a brief discussion of a refined and highly singular Hamiltonian which is obtained by supplementing H in (1.3.1) by a magnetic vector potential. This corresponds to the Schrödinger equation associated with a magnetic as well as an electric field. Further, in Example 11.4.10, we consider the case where a d-dimensional Riemannian manifold replaces Euclidean space \mathbb{R}^d .

Time-dependent potentials

The operator-valued Feynman integral used in Chapters 15–18 is defined via analytic continuation in mass. In those chapters, the emphasis is on Feynman's operational calculus and, in particular, on disentangling via time-ordered perturbation series by using the Wiener and Feynman integrals. The "potentials" allowed there are very general in most respects: they can be time-dependent and complex-valued and no smoothness assumptions are made. However, they are required to be essentially bounded in the space variables; that is, no spatial singularities are permitted. (Hence, for instance, the Coulomb potential is not allowed in this setting since it has a blow-up singularity at the origin.)

Potentials which are bounded and may be time-dependent appear in various places in the physics literature. Forces that are under the control of an experimenter provide a natural source of examples of potentials that are both time-dependent and bounded.

It is not just the potentials θ that influence the possible physical models in Chapters 15–18, but also the Lebesgue–Stieltjes measure η as in (1.1.4). These measures determine the disentangling (as noted earlier) and, when combined appropriately with an exponential function, determine the evolution of an associated physical system (see Chapter 17). [We refer, in particular, to Section 17.5 for possible physical interpretations of the corresponding results both in the quantum-mechanical (or Feynman) case and in the diffusion (or probabilistic) case.] The fact that such measures may have continuous and/or discrete parts allows us to study both continuous and discrete phenomena and their relationships with one another. This considerably broadens our approach to Feynman's operational calculus via Wiener and Feynman path integrals in Chapters 15–18. Mathematically, it also gives a rich combinatorial structure to the time-ordered perturbation expansions (or generalized Dyson series) and the associated generalized Feynman graphs introduced in Chapter 15 and used throughout the above chapters.

A brief discussion is given in Section 13.5 of Haugsby's extension of Nelson's second approach to the Feynman integral. This is the only place in the book where potentials are treated which can be both singular in the space variables and time-dependent.

Phenomenological models: complex and nonlocal potentials

We are also able to treat certain phenomenological models. By a phenomenological model, we mean one that does *not* arise from the basic principles of quantum mechanics

but has, nevertheless, been found useful in modeling certain quantum systems. We have already mentioned complex potentials above. Such potentials are used in modeling dissipative (or open) quantum systems. An extensive discussion of this topic—including its strengths and weaknesses and its relationship with "the" Feynman integral—can be found in Exner's book [Ex], Open Quantum Systems and the Feynman Integral. Complex potentials are permitted in some of the results in Chapters 11 and 13 (see especially Sections 11.6 and 13.6, as well as the end of Section 13.5) and in nearly all of the results in Chapters 15–18. The setting of Chapter 19 is more general, but operators of multiplication by a potential can be considered, and, when they are, the potentials involved can be both time-dependent and complex-valued.

Chapter 19 deals with time-dependent families $\{\beta(s): 0 \le s < \infty\}$ of bounded operators on a Hilbert space. (A strongly continuous semigroup of operators on the Hilbert space and the generator of that semigroup are also involved but are not particularly relevant to the present comments.) *Nonlocal* potentials are used phenomenologically in many body problems in several areas of quantum physics (see [Tab, ChSa, Mc] and the relevant references therein). The operator is an integral operator whose kernel V(x, y) (or V(s; x, y) if we have time-dependence) is referred to as a "nonlocal potential". It is nonlocal in that this "potential" does *not* depend on one sharp choice for the space coordinates (see formula (19.7.15)). Such nonlocal potentials are used, for example, in nuclear physics where the kernels used to model various situations are surprisingly simple; they are, in practice, separable kernels of finite (and low) rank (see Example 19.7.5).

Finally, we mention that some of the highly singular potentials discussed just above and treated in Section 11.4 and Sections 13.5-13.6 can also be viewed as providing suitable phenomenological models for certain problems occurring in quantum physics or in molecular chemistry. (See, for example, [LL, Ne1, FrLdSp].) For instance, the attractive inverse-square potential (Example 13.6.18) and more generally, highly singular attractive or repulsive central potentials (as in Examples 11.4.7 and 13.6.13), can be used to model problems occurring in quantum field theory or in polymer physics. They are often considered as "nonphysical" or only of academic interest because, in particular, they may lead (as in Example 13.6.18) to nonunitary evolutions and thus to Schrödinger operators which are no longer self-adjoint—in contradiction with one of the basic tenets of standard Hamiltonian quantum mechanics. (This unusual aspect is apprehended naturally within the context of the various approaches to analytic-in-mass Feynman integrals discussed in Sections 13.5 and 13.6; see [Ne1, Kat7, BivPi, BivLa].) Actually, the situation is somewhat more complicated than that and a suitable dose of pragmatism is needed to decide which model (whether of Feynman type or of Hamiltonian type, say) is most appropriate for a given physical situation; see, for instance, [Case, R, FrLdSp] and Example 13.6.18. In spite of these drawbacks, it can be argued convincingly that such highly singular potentials provide better approximate (or "phenomenological") models of suitable physical systems than their more regular counterparts. (See especially the review article [FrLdSp] as well as, for example, [LL, Ne1, PariZi, MarPari] and the relevant references therein.)

In closing the main part of this introduction, we briefly return to Chapter 20 which, as was mentioned earlier, is of a very different nature than the rest of this book. We

recall, in particular, that in Section 20.2, we discuss some of the relationships between heuristic Feynman-type integrals (as well as aspects of Feynman's operational calculus) and a variety of subjects from contemporary physics (or mathematics). In addition, in Section 20.2.A, several mathematical or physical models inspired by quantum field theory (specifically, "Chern-Simons gauge theory" defined in terms of a heuristic Feynman-Witten functional integral [Wit14]), are used to gain insight into (and extend) the celebrated Jones polynomial, along with other topological invariants that are central to modern knot theory and low-dimensional topology.

We hope that despite its relative brevity, Section 20.2 will prove helpful to a reader interested in getting a sense of the fascinating interplay between heuristic Feynman path integrals and a number of topics lying at the border of mathematics and theoretical physics.

Prerequisites, new material, and organization of the book

We end this introduction by making some specific comments about the content and the structure of this book, along the lines suggested in the title of the present italicized subsection.

As was mentioned in the preface and further explained earlier on in this chapter, much of the background material needed for the main part of this book is provided here; see especially Chapters 3.4, 6, 8–10, along with Chapter 7.

Detailed proofs—based mainly on the background material just referred to—are given for nearly all the theorems which deal with the main topics of this book. Most of the exceptions come in Sections 13.6, 13.7 and in the last part of Section 13.5, as well as in Section 17.6.

The reader will see that proofs are provided even for a good portion of the background material itself; see, in particular, Sections 3.1–3.4, 4.1–4.2, 4.5–4.6, and 10.2–10.3. We remark that if the reader is willing to forego the proofs in Sections 11.6, 13.5 and 13.6, then the operator-theoretic background needed for the book (especially through Chapter 13) is reduced to the information about self-adjoint operators and quadratic forms found in Section 9.6 and in Chapter 10 plus relatively few basic facts about semigroups of operators provided in Chapters 8 and 9.

We should mention that the comments in the preceding paragraphs do not apply to Chapter 20; no attempt there is made to supply proofs. (In the case of Section 20.2, in which much of the material connected with Feynman path integrals is of a heuristic nature, rigorous mathematical proofs are usually not known.)

The Lebesgue theory of measure and integration is employed in many places in this book. Precise references are typically given for the results used, but no systematic presentation of measure-theoretic prerequisites is provided. Brief discussions of this subject can be found in the books [BkExH, Appendix A, pp. 531–544], [Ru2, pp. 5–75] and [ReSi1, pp. 12–26]. Fuller treatments are given in many places, for example, in [Roy, Fol2, Coh, Du].

The list of references provided at the end of this book is extensive but certainly not complete. (We note that a significant fraction of the references is connected in some

way with Chapter 20, which deals with a broad selection of topics.) When the references are given in the main body of the text, they are typically presented in enough detail so that the relevant material can be easily located. The topics discussed in this book are interrelated in a variety of ways; we try to keep track of these relationships by systematic cross-referencing.

A substantial part of the material in this book other than the background material has appeared previously only in the research literature and, in a number of cases, only in the recent research literature. A few results that will play a prominent role come from sources that are not widely available. The primary example of that is the Sherbrooke monograph of the first author [Jo6] which plays a key role in the operator-valued analytic continuation in time results in Chapter 13.

There is a significant amount of novelty to the exposition in several places. For example, in Section 4.6, we discuss in detail what is meant by the "nonexistence of Feynman's measure" as well as related issues. Chapter 14 is an introduction to Feynman's operational calculus for noncommuting operators. This subject extends certain aspects of the Feynman integral, a fact that does not seem to be widely understood in the mathematical literature. We explain this in some detail in Chapter 14, and the idea is developed further in Chapters 15–19.

It will be clear to the reader of this book that the research interests of the authors have influenced much of the content. However, the influence went the other way as well; the desire to fill in missing pieces of the book directed some of our research in recent years. (For instance, reference [dFJoLa2]—on which Chapter 19 is based—was very much written with our book in mind.) A portion of that work is new here. We wish to call the reader's attention to a few such items.

In Section 13.4, we show that under rather general conditions, three of the four approaches that are discussed in this book are closely related. Most of this material is new.

The last part of Section 13.5 and nearly all of Section 13.6 deal with product formulas and operator-valued analytic continuation in mass from such formulas (rather than from the Wiener integral). A good portion of this material is new as well, particularly with regard to the explicit connections with the Feynman integral. Further, a comparison of the various analytic-in-mass Feynman integrals is provided, along with related results; see Theorems 13.6.10 and 13.6.11, along with Corollary 13.5.18. In addition, a detailed treatment of highly singular central potentials is given in this context; see Examples 13.6.13 and 13.6.18.

In Section 15.7, a "time-reversal map" is introduced and studied for our disentangling algebras in Chapters 15 and 18; see Definition 15.7.5, Theorem 15.7.6 and Corollary 15.7.8. This enables us, in particular, to clarify the connections with the usual "physical ordering" in the context of Feynman's operational calculus. In turn, these changes have repercussions in Chapter 17. We expect that more work will be done along related lines in the future.

New examples are provided in several places. We mention one, Example 15.5.3, that may be of particular interest. This example involves the purely continuous but singular measure associated with the Cantor function.

At the end of Chapter 16, we indicate how results from our earlier work (contained in [JoLa1]) on "stability" in the measures can be extended. We carry this out in detail in one case (see Proposition 16.2.14) and indicate how to go further in another case.

In Remark 11.5.15(d), we point out that the requirements on the negative part of the dominating "potential" in Theorem 11.5.13 (from [La12]) can be reduced to membership in the "Kato class" of functions on \mathbb{R}^d . (Theorem 11.5.13 is a dominated-type convergence theorem for the modified Feynman integral; it is the subject of Section 11.5 and is also applied in Chapter 13 to other approaches to the Feynman integral.)

Additional work on Feynman's operational calculus by Brian Jefferies and the first author ([JeJo]) is discussed briefly in III of Section 14.4. That work provides a nice supplement to the treatment given in Chapters 15–19 of Feynman's operational calculus (and based on [JoLa1-4. La14-18, dFJoLa1,2]). However, some aspects of the new material still need further development and so a fuller discussion of this topic could not be included in this book.

Several exercises or problems are proposed throughout the book. They are of varying degrees of difficulty. Typically, the exercises are mainly intended to illustrate a new concept, apply a new technique, or supplement some material in the text. Most of them should be accessible to graduate students. However, in a few instances, some of the proposed problems are extremely difficult and not yet solved in the literature (e.g. Problem 11.3.9). In other cases, they correspond to results already published but the proof of which is not discussed fully in the book (e.g. Problem 17.3.6 or 17.6.28). In addition, a few open-ended problems—the precise interpretation or formulation of which is left to the reader—are provided either formally (e.g. Problem 17.6.31) or in various comments or remarks scattered in the text. When appropriate, we have usually indicated the nature or the difficulty of the problem at hand.

The numbering system used in this book is straightforward. For example, Theorem 11.5.13 is the thirteenth numbered item in Section 5 of Chapter 11; a similar comment applies to the numbering of equations. Further, Section 15.4 is the fourth section of Chapter 15. Frequently, unnumbered subsections with italicized headings are used within a given section in order to delineate or highlight certain topics.

Indexes for symbols or notation, authors, and subjects are provided just after the bibliography. Along with the detailed list of contents, we hope that they will prove to be a useful guide to the reader throughout this book.